

Citation: Kennedy DA, Dwyer G (2018) Effects of multiple sources of genetic drift on pathogen variation within hosts. PLoS Biol 16(3): e2004444. https://doi.org/10.1371/journal.pbio.2004444

Academic Editor: Roland Regoes, ETH Zurich, Switzerland

Received: October 6, 2017

Accepted: February 26, 2018

Published: March 28, 2018

Copyright: © 2018 Kennedy, Dwyer. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Sequence data generated in this study are available through the Sequence Read Archive (SRA) of the National Center for Biotechnology Information (NCBI) under project accession number SRP107081. Authorgenerated code is available at GitHub repository: https://github.com/dkenned1/KennedyDwyer.

Funding: NIH (grant number R01-GM096655). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Effects of multiple sources of genetic drift on pathogen variation within hosts

David A. Kennedy^{1,2}, Greg Dwyer²*

1 Center for Infectious Disease Dynamics, Department of Biology, The Pennsylvania State University, University Park, Pennsylvania, United States of America, 2 Department of Ecology and Evolution, University of Chicago, Chicago, Illinois, United States of America

* gdwyer@uchicago.edu

Abstract

Changes in pathogen genetic variation within hosts alter the severity and spread of infectious diseases, with important implications for clinical disease and public health. Genetic drift may play a strong role in shaping pathogen variation, but analyses of drift in pathogens have oversimplified pathogen population dynamics, either by considering dynamics only at a single scale—such as within hosts or between hosts—or by making drastic simplifying assumptions, for example, that host immune systems can be ignored or that transmission bottlenecks are complete. Moreover, previous studies have used genetic data to infer the strength of genetic drift, whereas we test whether the genetic drift imposed by pathogen population processes can be used to explain genetic data. We first constructed and parameterized a mathematical model of gypsy moth baculovirus dynamics that allows genetic drift to act within and between hosts. We then quantified the genome-wide diversity of baculovirus populations within each of 143 field-collected gypsy moth larvae using Illumina sequencing. Finally, we determined whether the genetic drift imposed by host-pathogen population dynamics in our model explains the levels of pathogen diversity in our data. We found that when the model allows drift to act at multiple scales-including within hosts, between hosts, and between years-it can accurately reproduce the data, but when the effects of drift are simplified by neglecting transmission bottlenecks and stochastic variation in virus replication within hosts, the model fails. A de novo mutation model and a purifying selection model similarly fail to explain the data. Our results show that genetic drift can play a strong role in determining pathogen variation and that mathematical models that account for pathogen population growth at multiple scales of biological organization can be used to explain this variation.

Author summary

The genetic diversity of a pathogen population can strongly influence disease severity and spread. High genetic variation within hosts can lead to severe disease symptoms in individuals and rapid disease transmission in populations. But what factors determine this genetic diversity? Many factors might play roles, including natural selection, mutation,



Abbreviations: LdMNPV, *Lymantria dispar* multiple nucleopolyhedrovirus; NCBI, National Center for Biotechnology Information; SEIR, Suspectible-Exposed-Infectious-Removed; SRA, Sequence Read Archive.

and genetic drift (the randomness caused by chance events in small populations). The relative importance of these factors is unknown because diversity is presumably shaped by dynamics that occur at multiple scales (within and between hosts) and each scale is typically studied independently. We studied the combined impact of dynamics at multiple scales by constructing a mathematical model of pathogen dynamics that simultaneously describes both within-host and between-host disease dynamics. By combining scales, our approach allowed us to ask which factors most strongly influence pathogen genetic diversity. We compared our model to data from natural gypsy moth baculovirus infections. Although pathogen diversity is often assumed to be strongly shaped by natural selection, we found that genetic drift acting at multiple scales provided a better explanation for our data. This suggests that genetic drift plays a key role in shaping pathogen diversity within hosts, which in turn has implications for how we might better combat problematic pathogen evolution, such as the evolution of drug resistance.

Introduction

Pathogen genetic variation can have important consequences for human health in both clinical and epidemiological settings [1]. In particular, high variation within hosts can lead to severe disease symptoms within individuals and rapid disease transmission within populations [2, 3]. An understanding of the mechanisms determining pathogen variation might therefore lead to novel interventions, reducing the toll of infectious diseases. The development of such an understanding requires a quantification of the effects of population processes on pathogen genetic variation, in turn requiring mathematical models that relate population processes to genetic change.

Such models, however, tend to greatly simplify pathogen biology. Selection–mutation models, for example, often assume that pathogen populations are effectively infinite [4]. Models that allow pathogen population sizes to be finite typically neglect pathogen population processes, either within hosts in acute infections [5] or between hosts in chronic infections [6]. Models that attempt to capture both of these scales of disease dynamics have assumed either that pathogen population growth within hosts is very simple [7] or that pathogen bottlenecks at transmission are complete [7–9], so that every infection begins as a clonal lineage. These simplifications could strongly alter conclusions about the effects of genetic drift on pathogen diversity and indeed have been highlighted as key challenges in phylodynamic inference [10].

Genetic drift is a change in an allele's frequency due to the chance events that befall individuals. The effects of drift are therefore strongest in small populations, in which a few events can have a large impact [11]. The high population sizes typical of severe infections have led some authors to argue that drift has little effect on pathogens [12, 13], but pathogen population sizes typically fluctuate by several orders of magnitude over the course of infection and transmission. It therefore seems likely that, when pathogen populations are small and variable, pathogen genetic variation will be strongly affected by drift. Indeed, analyses that allow for finite population sizes have shown that drift has at least weak effects on some pathogens [6].

New infections are typically initiated by small pathogen population sizes within hosts [14], leading to bottlenecks at the time of transmission that may drive genetic drift. Pathogen population sizes within hosts can also remain small for long periods following exposure [15]. In small populations, chance events such as the timing of reproduction can strongly influence population growth, a phenomenon known as "demographic stochasticity" [16]. When the effects of demographic stochasticity are strong, chance may allow some virus strains to

replicate and survive while others go extinct, providing a second source of genetic drift that we refer to as "replicative drift." Note that we use the term "strain" to mean a population of pathogen particles that have identical genetic sequences.

Many previous studies of genetic drift in pathogens have focused only on population processes that operate within hosts, either during experiments with model organisms or during the treatment of human patients [14, 17]. Pathogen variation in nature, however, is also affected by processes that operate at the host population level, such as fluctuating infection rates during epidemics [18]. Studies of Ebola [19] and tuberculosis [20], for example, have shown that much of the variation present at the population level often cannot be explained by natural selection and must instead be due to neutral processes that presumably include genetic drift.

Genetic drift in pathogens may therefore be driven by population processes at multiple scales. These multiple scales can be incorporated into a single framework by constructing "nested" models, in which submodels of within-host pathogen population growth are nested in models of between-host pathogen transmission [21]. The computing resources necessary to analyze such complex models have only become available recently, however, and so it has not been clear whether sufficient data exist to test nested models of drift [22]. Indeed, even for models that assume that the strength of drift is constant across hosts, robust tests of the model predictions require both genetic data and mechanistic epidemiological models [9], a combination that is rarely available. Whether nested models can be of practical use for understanding pathogen genetic variation in nature is therefore unclear.

For baculovirus diseases of insects, pathogen population processes have been intensively studied at both the host population level [23] and at the individual host level [15]. Baculoviruses cause severe epizootics (epidemics in animals) in many insects [24], including economically important pest species such as the gypsy moth (*Lymantria dispar*) that we study here [25]. Collection and rearing protocols for the gypsy moth have long been standardized [26], and so previous studies of the gypsy moth baculovirus, *Lymantria dispar* multiple nucleopolyhedrovirus (LdMNPV), have produced parameter estimates for both within-host [27] and between-host [28–30] models. Moreover, collection of large numbers of virus-infected individuals is straightforward [25], making it possible to use high-throughput sequencing methods to characterize pathogen diversity across many virus-infected hosts. Here, we use a combination of whole-genome sequencing and parameterized, nested models to quantify the effects of genetic drift on the gypsy moth baculovirus. We show that a mechanistic model of genetic drift can explain variation in this pathogen but only if the model takes into account the effects of drift at multiple scales of biological organization.

Results

Sequencing the virus populations from each of 143 field-collected insects showed that there is substantial genetic variation in baculovirus populations between hosts. We generated consensus sequences for each of our 143 samples (Section A in <u>S1 Text</u>), and comparisons between consensus sequences identified 712 segregating sites at the between-host scale (defined as sites where alternative variants were the consensus in more than 6 samples [approximately equal to 5%]). These sites correspond to approximately 0.4% of the genome. Analysis of the variation at these 712 sites within each sampled virus population showed that these sites were polymorphic in some hosts but not others, which might occur if some hosts were exposed to multiple strains of virus while others were exposed to only a single strain. We summarize genetic variation within hosts using mean nucleotide diversity [<u>31</u>], the probability that 2 randomly selected alleles at a segregating site are different (Section A in <u>S1 Text</u>). Our conclusions were

unchanged when we used alternative metrics of diversity, such as the proportion of polymorphic loci, the effective number of alleles, or the relative nucleotide diversity (Section I in $\underline{S1}$ Text).

Calculating nucleotide diversity at the 712 segregating sites using the consensus sequences of our 143 samples showed that between-host nucleotide diversity $\pi = 0.404$. Within individual field-collected hosts, nucleotide diversity at these same sites ranged from 0.002 to 0.284 (mean = 0.072, SD = 0.077, Section B in S1 Text), while over all sites, nucleotide diversity ranged from 0.001 to 0.003, with a mean of 0.001. In Section B of S1 Text, we show that these values imply that a large fraction of nucleotide diversity within hosts can be explained by just 712 segregating sites, or 0.4% of the genome.

Together, these patterns suggest that substantial pathogen diversity within hosts is likely acquired from the exposure of host insects to multiple virus strains. If diversity had instead been generated by de novo mutation, nucleotide diversity between samples would have been less variable (Section E in <u>S1 Text</u>) and polymorphism would have likely been spread across many sites, including sites that were not polymorphic at the population level. Immune system-mediated diversifying selection is also an unlikely explanation because insects lack clonal immune cell expansion [<u>32</u>], because immune cell expansion does not explain why some hosts have substantially more pathogen diversity than others, and because we found no evidence of diversifying selection in our sequence data (Section H in <u>S1 Text</u>). Negative correlations between host families in susceptibility to different pathogen genotypes constitute yet a third unlikely explanation because such correlations are positive in the gypsy moth [<u>33</u>]. The migration of virus or infected larvae from nearby locations with different virus strains similarly cannot explain the data because population structure in the gypsy moth virus is minimal [<u>34</u>] (Section A in <u>S1 Text</u>).

Genetic drift, however, can explain the data but only if we allow for effects of population processes at multiple scales of biological organization. To explain why, we first use a nested model of pathogen population dynamics (Fig 1) to show how genetic drift in pathogen populations may operate at 3 scales: within hosts, within epizootics, and between years. We then show that the model can only explain the data if it includes effects of drift during both transmission bottlenecks and virus growth within hosts.

Simulations of our within-host model show that the combination of transmission bottlenecks and replicative drift can substantially reduce pathogen diversity within hosts (Fig 2A– 2C). Demographic stochasticity, which is manifest in the figure as jaggedness in the model trajectories, is strongest shortly after exposure, when the pathogen population size is small. This stochasticity generates variability in the time to host death, and it also drives replicative drift. Comparing this model to a linear birth–death model (Section C in S1 Text) shows that the immune system substantially slows the growth of the virus population early in the infection, which strengthens the effects of replicative drift.

Overwintered virus infects hatchlings during the initial emergence of hosts from eggs, an effect that is apparent in our simulations of the epizootic model (Fig 2D and 2E). After the overwintered virus decays, there is a short period when cadavers are rare, such that the vast majority of virus is present only within exposed larvae. When these exposed larvae die, the virus that they release is transmitted to new larvae feeding on foliage. During this time, the relative frequencies of different virus strains consumed by larvae can fluctuate strongly due to the drift that occurs when cadavers are rare. Low densities of cadavers can thus alter the relative frequency of strains within hosts. In the figure, the initial host population consists of more than 10,000 hosts, reflecting the high densities at which baculovirus epizootics occur in insect populations in nature [24]. Demographic stochasticity nevertheless influences the composition



Fig 1. Schematic of the nested model. Bottom: the host population size N_g and the infectious cadaver population size Z_g in generation g depend on host and pathogen population sizes in generation g - 1 and the disease dynamics in that generation. Following the epizootic, surviving hosts reproduce, and virus-killed cadavers overwinter to start the epizootic in the following year. Middle: the disease dynamics in generation g - 1 follow a stochastic SEIR model [35], such that a susceptible host S_i becomes exposed E_j to infectious cadaver P_k at rate $v_i q$, where v_i is the risk of exposure for host i and q is the probability of death given exposure, which arises from the within-host virus dynamics. Note that the "Removed" class R, corresponding to inactivated cadavers, is not explicitly shown. The probability of a host dying from virus infection at time τ post exposure, $p(\tau)$, is determined by the

dynamics of the pathogen within a host. *q* is related to $p(\tau)$ in that $q = \int_{0}^{\infty} p(\tau) d\tau$. Top: within a host, virus particles *x* can reproduce or interact with

immune cells *y*, resulting in the removal of both the virus particle and the immune cell. An infection fails to kill the host if all virus particles are cleared so that x = 0, but the host dies if the total number of virus particles reaches an upper threshold *C*. Further details are in Section C of <u>S1 Text</u>. To produce a model that lacks replicative drift, we assume that $f_l(\tau)$, the frequency of a virus strain *l* at time of death τ , is equal to the frequency of that strain immediately after the time of exposure $f_l(0)$. To produce a model that lacks transmission bottlenecks, we assume that the number of copies of a virus strain *l* at the beginning of an infection $x_l(0)$ is equal to the total number of virus particles that invade the host $\sum_l x_l(0)$ multiplied by the relative frequency of that virus strain in the cadaver that caused exposure $P_l / (\sum_p P_p)$. In the purifying selection model, host death occurs only if a larva was susceptible to one or more of the virus strains in the cadaver to which it was exposed. If so, the virus strains that the host was susceptible to are released upon host death at frequencies equal to those in the infecting cadaver. SEIR, Susceptible-Exposed-Infectious-Removed.

https://doi.org/10.1371/journal.pbio.2004444.g001

BIOLOGY

OS

of virus strains near the end of the epizootic, when the pathogen population begins to die out, in turn allowing drift to influence which virus strains cause infections within hosts.

Over longer time periods, fluctuations at the population scale (Fig 2F) produce host–pathogen cycles that match the dynamics of gypsy moth outbreaks in nature [29, 36]. These large fluctuations can drive changes in the relative frequency of pathogen strains, especially when pathogen population sizes and overall infection rates are at their lowest, in the troughs between



Fig 2. Simulations of the nested model. In all panels A–F, colored curves represent the pathogen population sizes of different virus strains, and the black curve shows the total pathogen population size. The colored bar at the top of each panel shows the relative frequencies of virus strains over time. Panels A–C show 3 realizations of the within-host virus growth model. A reexposure event, marked by a dashed, vertical red line, is also shown in panel C. The top colored panel left of time 0 shows the frequency of virus strains in a cadaver that a host was exposed to at time 0 (and reexposed to at time 50 in panel C). Death occurs when the total number of virus particles within a host hits an upper threshold. To aid visualization, here we set the pathogen population size at host death to be 10⁴, as opposed to the more realistic value of 10⁹ that we use when comparing our models to data. The time of death differs between simulations due to demographic stochasticity in virus growth, and in each simulation it is marked by a dashed, vertical black line. Panels D and E show 2 realizations of our stochastic SEIR-type epizootic model starting from identical initial conditions. Note that the curves here show cadaver quantities rather than virus particles as in panels A–C. Epizootics are initiated by overwintered cadavers that infect emerging larvae. As these cadavers decay, total cadaver quantity drops to low levels, such that the pathogen population is almost entirely composed of virus particles inside living hosts. These hosts then die, initiating future rounds of infections. Panel F shows a realization of our between-generation pathogen model, with trajectories showing the total number of virus-killed hosts in each generation. The frequency of pathogen strains can drift over time, an effect that is particularly noticeable during troughs of infection. SEIR, Susceptible-Exposed-Infectious-Removed.

https://doi.org/10.1371/journal.pbio.2004444.g002

host population peaks. Host-pathogen population cycles in our model thus further strengthen the effects of genetic drift on the pathogen.

Our combined model shows that drift can act both within and between hosts and at timescales ranging from hours to decades. One way to test the model would therefore be to collect data on pathogen genotypes over time, to see whether observed changes in genotype frequencies match those predicted by the model. However, logistical constraints common across many infectious disease systems, such as the long time intervals between epizootics and the difficulty of collecting samples when infection is rare, precluded the collection of such extensive data. We therefore instead used our model to predict the expected distribution of nucleotide diversity across hosts during an epizootic, and we compared this distribution to the corresponding distribution from our sequence data.

To test whether the data can be explained equally well by models that neglect one or more sources of drift, we also tested models that eliminated replicative drift or that eliminated both replicative drift and transmission bottlenecks. Note that it is not possible to construct a model that includes replicative drift but not transmission bottlenecks because replicative drift requires that virus population sizes be integer values, and forcing the virus population to be an integer value necessarily imposes a form of bottleneck. Also, to test whether the data are better explained by selection than by drift, we constructed a model that allows for purifying selection to act within hosts but that lacks both replicative drift and transmission bottlenecks.

These comparisons show that only the model that includes both replicative drift and bottlenecks can explain the data (Fig 3). The model that includes only drift at the host population scale predicts within-host diversity levels that are much higher and much less variable than in the data. The model that includes population scale drift and bottlenecks but not replicative drift as well as the model that includes purifying selection but not transmission bottlenecks or replicative drift both correctly predict that there will be substantial variation across hosts, but they predict diversity levels that are much higher than in the data. In contrast, the model that includes replicative drift and transmission bottlenecks accurately predicts the entire distribution of diversity levels seen in the data. This visual impression is confirmed by differences in



Fig 3. Fit of models to data. Comparison of the predictions of our models (gray-shaded areas, showing 95% CIs of model realizations) to the distribution of nucleotide diversity within 143 individual infected hosts calculated from our sequence data (black dots show data on nucleotide diversity within hosts). Panel A shows the predictions of a model that lacks both transmission bottlenecks and replicative drift, panel B shows the predictions of a model that includes transmission bottlenecks but not replicative drift, panel C shows the predictions of a model that includes both transmission bottlenecks and replicative drift, and panel D shows the predictions of a model that includes purifying selection within hosts but not transmission bottlenecks or replicative drift. Values underlying data points are provided in S1 Data.

https://doi.org/10.1371/journal.pbio.2004444.g003

the Monte Carlo estimates of the likelihood scores across models (Section F in S1 Text: neutral model with neither bottlenecks nor replicative drift, median log mean likelihood = -503.2; purifying selection model, median log mean likelihood = -353.1; neutral model with bottlenecks but not replicative drift, median log mean likelihood = -266.7; neutral model with both bottlenecks and replicative drift, median log mean likelihood = -63.9). Because no parameters were fit to the diversity data, we do not need a model complexity penalty, but the difference in the number of parameters across models was, in any case, dwarfed by the differences in the likelihood scores. In Section I of S1 Text, we show that the differences in model likelihoods hold when using alternative diversity metrics, and in Sections D and G of S1 Text, we show that our qualitative conclusions are robust across parameter values that determine bottleneck severity and selection intensity.

Our results thus show that a model that accounts for the effects of population processes at multiple scales can explain differences in pathogen variation across hosts in the gypsy moth baculovirus. In contrast, models that simplify the effects of genetic drift by ignoring effects of transmission bottlenecks and replicative drift, or that allow for selection but not within-host drift, cannot explain the diversity of this pathogen. More broadly, because the model parameters were estimated entirely from experimental data on baculovirus infection rates (Section C in S1 Text), we are effectively carrying out cross-validation of the model.

The highly skewed distribution of nucleotide diversity apparent in our data can therefore be explained by a model that allows for drift at multiple scales and that includes multiple sources of drift within hosts but not by simpler models. In addition, Fig 4 shows that the best model can reproduce entire distributions of diversity within individual hosts. Allele frequency distributions in the model nevertheless tend to have slightly shorter tails and narrower peaks compared to the data. These mild discrepancies may be partially explained by mutations that occurred during viral passaging or during library preparation, but they can also be explained by small biases introduced during the mapping of our short sequence reads to the reference genome (Section J in S1 Text). The data therefore do not reject the model.



Fig 4. Allele frequencies within hosts. Representative distributions of allele frequencies from individual hosts in our best model (A–E) and in our data (F–J). Each plot shows the distribution of allele frequencies within a single individual at 712 segregating sites, showing only the frequency of the most common allelic variant at each locus within that host. The number on each plot is the mean nucleotide diversity within that particular host. Model plots are aligned with similar data plots. The lack of diversity in panels A and F suggests that the virus population within these hosts consists of only a single virus strain. The bimodal distributions in panels B, C, and G suggest that these virus populations contain exactly 2 virus strains. The high diversity but lack of bimodality in panels D, E, H, I, and J suggests that these virus populations consist of more than 2 virus strains. Values underlying panels F–J are provided in S2 Data.

https://doi.org/10.1371/journal.pbio.2004444.g004





Fig 5. Dynamics over time. Model predictions of the effects of changes in the populations of susceptible and infected hosts on within-host pathogen diversity, over the host–pathogen population cycle. (A) The population size of uninfected hosts. (B) The population size of infectious cadavers (blue) and the mean nucleotide diversity (red).

https://doi.org/10.1371/journal.pbio.2004444.g005

Our virus samples were collected at times of peak or near-peak gypsy moth densities, which are the only times at which large numbers of larvae can be collected easily, and so the data do not directly show how changes in pathogen population size at the host population scale affect pathogen variation. We therefore used our best model to explore how pathogen variation within hosts will change over the course of the gypsy moth outbreak cycle. Within-host diversity is predicted to be highest just as the host population begins to crash due to the pathogen, after which diversity is predicted to gradually decline until the next outbreak (Fig 5). Reductions in within-host variation due to transmission bottlenecks and replicative drift are thus counterbalanced by increases in within-host variation at the time of host population peaks, due to the high frequency of multiple exposures when host populations are large. Long-term, population scale processes can therefore also strongly affect within-host variation.

Discussion

A basic prediction of population genetics theory [11] and a fundamental assumption of phylodynamic modeling [18] is that the effects of genetic drift are determined by population processes. Explicit tests of this assumption for infectious diseases, however, are rare. We used a model that was developed and parameterized using nongenetic datasets to show that patterns of genetic diversity in an insect pathogen can be explained by a model that accounts for population processes at multiple scales but not by models that simplify or neglect the effects of drift. Previous work has attempted to infer disease demography and pathogen evolution from genetic data [18]. Here, we instead began with an existing population process model that has already been fit to epidemiological data, and we used it to predict pathogen genetic data. We therefore tested the extent to which disease demography can be used to predict neutral pathogen evolution. By using this approach, we have shown that evolutionary hypotheses can be tested by integrating genetic data and ecological models. A simple model of purifying selection was not able to explain the patterns of diversity in our data. Models that instead invoke diversifying selection or more complex patterns of host-specific immune selection might provide reasonable explanations for our data, but such models require extra parameters to account for the costs and benefits of alternative alleles, increasing the complexity of the models [37]. Moreover, drift is an inherent property of small populations, and so models that invoke selection should still allow for effects of drift if population sizes are small. In our case, complex models of selection were not needed to explain patterns of diversity, suggesting that the effects of selection on our data are weak relative to the effects of drift. Selection may nevertheless be necessary to explain variation in other pathogens or other datasets. Given that polymorphism has been widely observed in insect baculoviruses [38, 39], our results suggest that baculoviruses present opportunities to understand the relationship between host–pathogen ecology and pathogen diversity.

We have shown that both transmission bottlenecks and replicative drift have detectable impacts on pathogen diversity. Due to the difficulty of separating these effects, previous studies of genetic drift have assumed that bottlenecks are complete [6–8], have ignored impacts of key biological processes such as the host immune response [40], or have summarized the effects of multiple sources of drift with a single parameter—the effective population size N_e [41]. Similarly, estimates of bottleneck size often combine the effects of transmission bottlenecks and replicative drift into a single estimate of the effective bottleneck, biasing estimates of transmission bottleneck size (section D in S1 Text) [40]. Distinguishing between transmission bottlenecks and replicative drift, however, may provide novel insights into disease control strategies. For example, the emergence of resistance to antibiotic drugs might be slowed if drug therapy windows are restricted to periods when the effects of replicative drift are strongest, such as when pathogen populations are small or are turning over rapidly.

To show that both transmission bottlenecks and replicative drift play an important role in shaping pathogen diversity within hosts, we have focused our analysis on common variants that cannot be easily explained by de novo mutation. Additional variation is nevertheless present (Section B in S1 Text). In our case, this other variation occurs at such low levels that it cannot be readily distinguished from sequencing error, but it is almost certainly true that mutation and selection also play roles in shaping total pathogen diversity within hosts. Our argument is therefore not that mutation and selection are unimportant but instead that transmission bottlenecks and replicative drift can strongly affect pathogen diversity within hosts. In our case, bottlenecks and replicative drift appear to be the main drivers of diversity at sites that segregate at the population level.

High-throughput sequencing has revolutionized our ability to measure pathogen variation. It has been used to detect drug resistance [42], to discover novel viruses in nature [43], and to diagnose disease in clinical settings [44]. Our work shows that high-throughput sequencing can also provide important insights into the ecology and evolution of host-pathogen interactions, especially when combined with nested disease models. The increasing availability of both parameterized models [35] and genomic data [45] suggests that our approach of using genetic data to challenge models of nested disease dynamics may be widely applicable.

Materials and methods

Ethics statement

This research was conducted under United States Department of Agriculture Animal and Plant Health Inspection Service permit P526P-12-01466.

Model description

The gypsy moth baculovirus LdMNPV is a double-stranded DNA virus with an approximately 161-kb genome. The virus belongs to the family Baculoviridae, and—like all baculoviruses—it exists in 2 forms, as an "occlusion body" that is highly stable in the environment due to its protective proteinaceous matrix and as a "budded virus" that is released from cells during replication within hosts.

The gypsy moth baculovirus is transmitted when larvae consume occlusion bodies while feeding on foliage [29]. If the resulting virus population grows inside the host to a sufficiently large size, the larva dies, releasing new occlusion bodies onto the foliage. These occlusion bodies are then available to be consumed by additional conspecifics (the virus is species specific [24]), leading to very high infection rates in high-density populations [25]. During the fall and winter, when the insect is in the egg stage, the virus persists in locations where the virus may be protected from degradation by ultraviolet light such as beneath egg masses laid on cadavers [46, 47]. Genetic drift in the gypsy moth baculovirus may therefore be affected by population processes at multiple scales, including within individual hosts and across the host population.

Exposure to the virus results in an initial population of only a few virus particles [15, 48], and the population size in the host remains small for a substantial period of time following exposure [15, 27]. Our model of pathogen growth within hosts therefore tracks population sizes from the initial population bottleneck through the stochastic growth of the pathogen population, until death or recovery. Our model thus explicitly includes genetic drift (Fig 1).

Our within-host model is based on a birth-death model [16], which describes probabilistic changes in population sizes over time. In birth-death models, the probability of a birth or a death in a small period of time increases with the population size [49]. When the population size is small in a birth-death model, it is possible for extinction to occur due to a chance preponderance of deaths over births, even if the per capita birth rate exceeds the per capita death rate. Birth-death models are therefore well suited to describe the demographic stochasticity that underlies replicative drift.

In our within-host birth-death model, pathogen extinction is equivalent to the clearance of the infection by the host. If the pathogen does not go extinct, its population eventually becomes large enough that the effects of stochasticity are negligible [50], leading to host death when the population reaches an upper threshold. In previous work, we showed that linear birth-death models are insufficient to explain data on the speed of kill of the gypsy moth baculovirus, whereas models that allow for nonlinearities due to the immune system produce a better explanation for the data [15].

Our within-host model therefore describes virus removal as the outcome of a process that begins with the insect's immune system releasing chemicals that activate the phenol-oxidase pathway. This release causes virus particles to be encapsulated and destroyed by host immune cells, and it also incapacitates the immune cell [51, 52]. Our model therefore follows standard predator-prey-type immune system models [53], in which the pathogen is the prey and the immune cells are the predator—except that here, the immune cells do not reproduce over the timescale of a single infection. The pathogen population in the model may then be driven to 0 because of interactions with the host immune system, or it may persist long enough to overwhelm the host immune system, leading to exponential pathogen population size and on demographic stochasticity during the infection.

In our model, the initial pathogen population size within a host is drawn from a Poisson distribution (Fig 1) [15]. If the infecting cadaver is composed of multiple strains, the model draws an initial population size for each strain from a multinomial distribution, such that the

probability of sampling a particular strain from the infecting cadaver depends on the frequency of that strain in the cadaver. This process creates a transmission bottleneck. Next, the model tracks the population size of each virus strain over the course of the infection. Changes in the relative frequencies of these strains over time creates replicative drift. The host dies when the total pathogen population size exceeds an upper threshold. The frequency of virus strains at the time of host death determines the frequency of strains in the newly generated cadaver.

We model pathogen dynamics at the scale of the entire host population first by using a stochastic Susceptible-Exposed-Infectious-Removed (SEIR) model to describe epizootics (in our case, the infected *I* class consists of infectious cadavers in the environment, which we symbolize as *P* for pathogen). Our SEIR model is modified to allow hosts to vary in infection risk, an important feature of gypsy moth virus transmission [29, 54], and to allow exposed hosts to be reexposed because infected gypsy moth larvae continue to consume foliage until shortly before death [55]. For computational convenience [56], most SEIR models assume that the time between exposure and infectiousness follows a gamma distribution [35]. We instead allow this time to be determined by our within-host model so that the within-host model is nested inside the stochastic SEIR model. As in the within-host model, the frequency of different virus strains at the population scale can drift due to chance events, such as the exposure of hosts to one cadaver and not another. Our between-host model therefore adds an additional source of drift to our nested models.

Over longer timescales, gypsy moth populations go through host–pathogen population cycles, in which host outbreaks are terminated by baculovirus epizootics. This pattern is typical of many forest-defoliating insects [24]. The resulting predator-prey–type oscillations drive gypsy moth outbreaks at intervals of 5 to 9 years (we neglect the effects of the gypsy moth fungal pathogen *Entomophaga maimaiga*, which had only modest effects in our study areas in Michigan, US, in the years when we collected our samples) [57]. Virus infection rates are very low between insect outbreaks [29], which may strengthen the effects of genetic drift.

Gypsy moths have only 1 generation per year and thus only 1 epizootic per year. We therefore nest our within-host/SEIR-type model into a model that describes host reproduction and virus survival after the epizootic (Fig 1). The SEIR model determines which hosts die during the epizootic and which virus strains killed those hosts. This information is used in difference equations that describe the reproduction of the surviving hosts, the survival of the pathogen over the winter, and the evolution of host resistance, an important factor in gypsy moth outbreak cycles [29, 36].

By explicitly tracking the dynamics of individual hosts and pathogens, our model inherently includes the effects of genetic drift. We also tested whether a simple model of purifying selection or a model of de novo mutation could explain the patterns of diversity in the data without invoking drift within hosts. If these models were to fail to explain the patterns of diversity seen in our data, more complex models of evolution would need to be considered. In the gypsy moth baculovirus system, however, mutation rates are likely low [58, 59], spatial structure appears to be weak [34] (Section A in S1 Text), and evidence of selection acting within hosts is lacking (Section H in S1 Text). Drift therefore seems likely to play a strong role in shaping virus diversity.

To show that the different sources of drift in our model are actually necessary to explain the data, we created 3 alternative models. All 3 alternative models simplify the effects of genetic drift, but 1 also allows for effects of purifying selection. For the first alternative model, we simplified the effects of genetic drift by assuming that the relative frequencies of different virus strains within hosts do not change during pathogen population growth within hosts. To do this, we altered the model output such that the relative frequencies of virus strains released

from a host upon host death were equal to the relative frequencies of virus strains just after the transmission bottleneck, thereby eliminating the effects of replicative drift (Fig 1). For the second alternative model, we further simplified the effects of drift by assuming that the relative frequencies of virus strains at the end of an infection were the same as their relative frequencies in the infectious cadaver that initiated the infection, thereby eliminating both replicative drift and transmission bottlenecks (Fig 1). For the third alternative model, we added purifying selection to the second alternative model, which lacked both replicative drift and transmission bottlenecks, by assuming that each host was susceptible to only a random subset of virus strains. Exposure would therefore only result in death if a host was susceptible to one or more virus strains in the cadaver to which it was exposed. The relative frequencies of virus strains to which that host was susceptible in the infecting cadaver.

Baculovirus sequencing

We collected larvae from outbreaking gypsy moth populations in Michigan between 2000 and 2003 (Section A in <u>S1 Text</u>), and we reared the larvae until they pupated or died of infection [25]. The virus population from each virus-killed larva was passaged once by infecting 75 larvae with liquefied cadaver to generate enough virus for DNA extraction. We then extracted DNA following a standard baculovirus DNA extraction protocol, and we amplified the DNA using whole-genome amplification (REPLI-g UltraFast Mini kit from Qiagen).

We constructed sequencing libraries using the Nextera DNA Sample Prep Kit (Illuminacompatible, #GA0911-96) with custom barcodes to distinguish between the virus communities of different hosts. Our barcodes consisted of the first 96 indexes proposed by reference [60] (Section A in S1 Text). Sequencing was carried out as 2 sets of libraries, run on individual lanes of a HiSeq2000 at the University of Illinois at Urbana–Champaign, producing 100-cycle single-end reads. Samples were separated by barcode using the standard Illumina pipeline, and adaptor contamination was removed using "trim_galore." Reads were mapped to the firstsequenced gypsy moth baculovirus genome [61] using "bowtie2" [62] with the parameter set "very fast" (Section A in S1 Text). Overall mean coverage was 886x and varied across samples from 202x to 1,497x (Section A in S1 Text). Variant calling was carried out using VarScan version 2.3.9 [63]. More details can be found in Section A of S1 Text.

Supporting information

S1 Text. Supplemental information. (PDF)

S1 Data. Observed mean nucleotide diversities. (CSV)

S2 Data. Major allele frequencies shown in Fig 4. (CSV)

S3 Data. Sample names, collection sites, and collection years. (CSV)

S4 Data. Consensus sequences. Columns show data for different loci. Rows show data for different samples. Single dots indicate loci where the consensus sequence matched the reference genome.

(CSV)

S5 Data. *F*_{ST} values. (CSV)

S6 Data. Mean genome coverage for each sample. (CSV)

S7 Data. Mean sequencing depth for each locus. (CSV)

S8 Data. Percent of reads mapping to LdMNPV in BLAST search. LdMNPV, *Lymantria dispar* multiple nucleopolyhedrovirus.

(CSV)

S9 Data. Observed effective number of alleles. (CSV)

S10 Data. Observed proportion of polymorphic sites. (CSV)

S11 Data. Observed relative nucleotide diversities. (CSV)

S12 Data. Major allele frequencies in simulated single-infected host. (CSV)

S13 Data. Major allele frequencies in simulated multiple-infected host. (CSV)

Acknowledgments

We thank S. Allesina, J. Bergelson, M. Kreitman, V. Morley, and C. Pfister for comments on earlier versions of the text. Computational support was provided by the Computing Research Institute and the Research Computing Center at the University of Chicago.

Author Contributions

Conceptualization: David A. Kennedy, Greg Dwyer. Data curation: David A. Kennedy. Formal analysis: David A. Kennedy. Funding acquisition: Greg Dwyer. Methodology: David A. Kennedy. Resources: Greg Dwyer. Software: David A. Kennedy. Supervision: Greg Dwyer. Validation: David A. Kennedy. Visualization: David A. Kennedy, Greg Dwyer. Writing – original draft: David A. Kennedy, Greg Dwyer.

References

- 1. Alizon S, Luciani F, Regoes RR. Epidemiological and clinical consequences of within-host evolution. Trends Microbiol. 2011; 19(1):24–32. https://doi.org/10.1016/j.tim.2010.09.005 PMID: 21055948
- Read AF, Taylor LH. The ecology of genetically diverse infections. Science. 2001; 292(5519):1099– 1102. PMID: 11352063
- Vignuzzi M, Stone JK, Arnold JJ, Cameron CE, Andino R. Quasispecies diversity determines pathogenesis through cooperative interactions in a viral population. Nature. 2006; 439(7074):344–348. https://doi.org/10.1038/nature04388 PMID: 16327776
- Lorenzo-Redondo R, Fryer HR, Bedford T, Kim EY, Archer J, Pond SLK, et al. Persistent HIV-1 replication maintains the tissue reservoir during therapy. Nature. 2016; 530(7588):51+. https://doi.org/10. 1038/nature16933 PMID: 26814962
- Koelle K, Cobey S, Grenfell B, Pascual M. Epochal evolution shapes the phylodynamics of interpandemic influenza A (H3N2) in humans. Science. 2006; 314(5807):1898–1903. <u>https://doi.org/10.1126/</u> science.1132745 PMID: 17185596
- Pennings PS, Kryazhimskiy S, Wakeley J. Loss and recovery of genetic diversity in adapting populations of HIV. PLoS Genet. 2014; 10(1):e1004000. https://doi.org/10.1371/journal.pgen.1004000 PMID: 24465214
- Klinkenberg D, Backer JA, Didelot X, Colijn C, Wallinga J. Simultaneous inference of phylogenetic and transmission trees in infectious disease outbreaks. PLoS Comput Biol. 2017; 13(5):e1005495. <u>https:// doi.org/10.1371/journal.pcbi.1005495</u> PMID: 28545083
- Ypma RJ, van Ballegooijen WM, Wallinga J. Relating phylogenetic trees to transmission trees of infectious disease outbreaks. Genetics. 2013; 195(3):1055–1062. <u>https://doi.org/10.1534/genetics.113</u>. 154856 PMID: 24037268
- Didelot X, Gardy J, Colijn C. Bayesian inference of infectious disease transmission from whole-genome sequence data. Mol Biol Evol. 2014; 31(7):1869–1879. https://doi.org/10.1093/molbev/msu121 PMID: 24714079
- Frost SD, Pybus OG, Gog JR, Viboud C, Bonhoeffer S, Bedford T. Eight challenges in phylodynamic inference. Epidemics. 2015; 10:88–92. https://doi.org/10.1016/j.epidem.2014.09.001 PMID: 25843391
- 11. Nagylaki T. Introduction to Theoretical Population Genetics. Berlin Heidelberg: Springer-Verlag; 1992.
- Kouyos RD, Althaus CL, Bonhoeffer S. Stochastic or deterministic: what is the effective population size of HIV-1? Trends Microbiol. 2006; 14(12):507–511. https://doi.org/10.1016/j.tim.2006.10.001 PMID: 17049239
- Maldarelli F, Kearney M, Palmer S, Stephens R, Mican J, Polis MA, et al. HIV populations are large and accumulate high genetic diversity in a nonlinear fashion. J Virol. 2013; 87(18):10313–10323. <u>https://doi.org/10.1128/JVI.01225-12 PMID: 23678164</u>
- Gutiérrez S, Michalakis Y, Blanc S. Virus population bottlenecks during within-host progression and host-to-host transmission. Curr Opin Virol. 2012; 2(5):546–555. <u>https://doi.org/10.1016/j.coviro.2012</u>. 08.001 PMID: 22921636
- Kennedy DA, Dukic V, Dwyer G. Pathogen growth in insect hosts: inferring the importance of different mechanisms using stochastic models and response-time data. Am Nat. 2014; 184(3):407–423. https://doi.org/10.1086/677308 PMID: 25141148
- 16. Kot M. Elements of Mathematical Ecology. Cambridge: Cambridge University Press; 2001.
- Abel S, zur Wiesch PA, Davis BM, Waldor MK. Analysis of bottlenecks in experimental models of infection. PLoS Pathog. 2015; 11(6):e1004823. <u>https://doi.org/10.1371/journal.ppat.1004823</u> PMID: 26066486
- Grenfell BT, Pybus OG, Gog JR, Wood JLN, Daly JM, Mumford JA, et al. Unifying the epidemiological and evolutionary dynamics of pathogens. Science. 2004; 303(5656):327–332. https://doi.org/10.1126/ science.1090727 PMID: 14726583
- Azarian T, Presti AL, Giovanetti M, Cella E, Rife B, Lai A, et al. Impact of spatial dispersion, evolution, and selection on Ebola Zaire Virus epidemic waves. Sci Rep. 2015; 5.
- Lee RS, Radomski N, Proulx JF, Levade I, Shapiro BJ, McIntosh F, et al. Population genomics of Mycobacterium tuberculosis in the Inuit. Proc Natl Acad Sci USA. 2015; 112(44):13609–13614. <u>https://doi.org/10.1073/pnas.1507071112</u> PMID: 26483462
- 21. Mideo N, Alizon S, Day T. Linking within-and between-host dynamics in the evolutionary epidemiology of infectious diseases. Trends Ecol Evol. 2008; 23(9):511–517. https://doi.org/10.1016/j.tree.2008.05. 009 PMID: 18657880

- Gog JR, Pellis L, Wood JL, McLean AR, Arinaminpathy N, Lloyd-Smith JO. Seven challenges in modeling pathogen dynamics within-host and across scales. Epidemics. 2015; 10:45–48. https://doi.org/10. 1016/j.epidem.2014.09.009 PMID: 25843382
- Elderd BD. Developing models of disease transmission: insights from ecological studies of insects and their baculoviruses. PLoS Pathog. 2013; 9(6):e1003372. https://doi.org/10.1371/journal.ppat.1003372 PMID: 23785277
- Moreau G, Lucarotti CJ. A brief review of the past use of baculoviruses for the management of eruptive forest defoliators and recent developments on a sawfly virus in Canada. Forest Chron. 2007; 83(1):105– 112.
- **25.** Woods SA, Elkinton JS. Bimodal patterns of mortality from nuclear polyhedrosis-virus in gypsy-moth (*Lymantria-dispar*) populations. J Invertebr Pathol. 1987; 50:151–157.
- Elkinton JS, Liebhold AM. Population dynamics of gypsy moth in North America. Annu Rev Entomol. 1990; 35:571–596.
- Kennedy DA, Dukic V, Dwyer G. Combining principal component analysis with parameter line-searches to improve the efficacy of Metropolis–Hastings MCMC. Environ Ecol Stat. 2015; 22(2):247–274.
- 28. Fuller E, Elderd BD, Dwyer G. Pathogen persistence in the environment and insect-baculovirus interactions: disease-density thresholds, epidemic burnout and insect outbreaks. Am Nat. 2012; 179(3).
- Elderd BD, Dushoff J, Dwyer G. Host-Pathogen Interactions, Insect Outbreaks, and Natural Selection for Disease Resistance. Am Nat. 2008; 172:829–842. https://doi.org/10.1086/592403 PMID: 18976065
- Elderd BD, Rehill BJ, Haynes KJ, Dwyer G. Induced plant defenses, host–pathogen interactions, and forest insect outbreaks. P Natl Acad Sci USA. 2013; 110(37):14978–14983.
- **31.** Nei M, Li WH. Mathematical model for studying genetic variation in terms of restriction endonucleases. Proc Natl Acad Sci USA. 1979; 76(10):5269–5273. PMID: 291943
- 32. Vilmos P, Kurucz E. Insect immunity: Evolutionary roots of the mammalian innate immune system. Immunol Lett. 1998; 62:59–66. PMID: 9698099
- Hudson AI, Fleming-Davies AE, Páez DJ, Dwyer G. Genotype-by-genotype interactions between an insect and its pathogen. J Evolution Biol. 2016; 29(12):2480–2490.
- **34.** Fujita PA. Combining models with empirical data to examine dispersal mechanisms in the gypsy moth nucleopolyhedrosis host-pathogen system [Ph.D. Dissertation]. University of Chicago; 2007.
- 35. Keeling MJ, Rohani P. Modeling Infectious Diseases. New Jersey: Princeton University Press; 2008.
- Dwyer G, Dushoff J, Elkinton JS, Levin SA. Pathogen-driven outbreaks in forest defoliators revisited: Building models from experimental data. Am Nat. 2000; 156(2):105–120. https://doi.org/10.1086/ 303379 PMID: 10856195
- Orr HA. Testing natural selection vs. genetic drift in phenotypic evolution using quantitative trait locus data. Genetics. 1998; 149(4):2099–2104. PMID: 9691061
- Chateigner A, Bézier A, Labrousse C, Jiolle D, Barbe V, Herniou EA. Ultra deep sequencing of a baculovirus population reveals widespread genomic variations. Viruses. 2015; 7(7):3625–3646. <u>https://doi.org/10.3390/v7072788 PMID: 26198241</u>
- Hodgson DJ, Vanbergen AJ, Watt AD, Hails RS, Cory JS. Phenotypic variation between naturally coexisting genotypes of a Lepidopteran baculovirus. Evol Ecol Res. 2001; 3:687–701.
- Sobel Leonard A, Weissman D, Greenbaum B, Ghedin E, Koelle K. Transmission Bottleneck Size Estimation from Pathogen Deep-Sequencing Data, with an Application to Human Influenza A Virus. J Virol. 2017; p. JVI–00171.
- Volz EM, Romero-Severson E, Leitner T. Phylodynamic Inference across Epidemic Scales. Mol Biol Evol. 2017; 34(5):1276–1288. https://doi.org/10.1093/molbev/msx077 PMID: 28204593
- 42. Mideo N, Bailey JA, Hathaway NJ, Ngasala B, Saunders DL, Lon C, et al. A deep sequencing tool for partitioning clearance rates following antimalarial treatment in polyclonal infections. Evol Med Public Health. 2016; 2016(1):21–36. https://doi.org/10.1093/emph/eov036 PMID: 26817485
- 43. Lipkin WI, Anthony SJ. Virus hunting. Virology. 2015; 479:194–199. https://doi.org/10.1016/j.virol.2015. 02.006 PMID: 25731958
- Wilson MR, Naccache SN, Samayoa E, Biagtan M, Bashir H, Yu G, et al. Actionable diagnosis of neuroleptospirosis by next-generation sequencing. New Engl J Med. 2014; 370(25):2408–2417. <u>https://doi.org/10.1056/NEJMoa1401268 PMID: 24896819</u>
- Hatherell HA, Colijn C, Stagg HR, Jackson C, Winter JR, Abubakar I. Interpreting whole genome sequencing for investigating tuberculosis transmission: a systematic review. BMC Med. 2016; 14(1):1.
- Murray KD, Elkinton JS. Environmental contamination of egg masses as a major component of transgenerational transmission of gypsy-moth nuclear polyhedrosis-virus (LdMNPV). J Invertebr Pathol. 1989; 53(3):324–334.

- Fleming-Davies AE, Dwyer G. Phenotypic Variation in Overwinter Environmental Transmission of a Baculovirus and the Cost of Virulence. Am Nat. 2015; 186(6):797–806. <u>https://doi.org/10.1086/683798</u> PMID: 26655986
- Zwart MP, Hemerik L, Cory JS, de Visser JAGM, Bianchi FJJA, Van Oers MM, et al. An experimental test of the independent action hypothesis in virus-insect pathosystems. Proc R Soc Lond B. 2009; 276(1665):2233–2242.
- **49.** Renshaw E. Modeling Biological Populations in Space and Time. Cambridge: Cambridge University Press; 1991.
- Saaty TL. Some stochastic-processes with absorbing barriers. J R Stat Soc Series B Stat Methodol. 1961; 23:319–334.
- Ashida, Brey PT. In: Brey PT, Hultmark D, editors. Molecular Mechanisms of Immune Responses in Insects. London: Chapman & Hall; 1998.
- Trudeau D, Washburn JO, Volkman LE. Central role of hemocytes in Autographa californica M nucleopolyhedrovirus pathogenesis in Heliothis virescens and Helicoverpa zea. J Virol. 2001; 75(2):996– 1003. https://doi.org/10.1128/JVI.75.2.996-1003.2001 PMID: 11134313
- Alizon S, van Baalen M. Acute or chronic? Within-host models with immune dynamics, infection outcome and parasite evolution. Am Nat. 2008; 172(6):E244–E256. <u>https://doi.org/10.1086/592404</u> PMID: 18999939
- 54. Dwyer G, Elkinton JS, Buonaccorsi JP. Host heterogeneity in susceptibility and disease dynamics: Tests of a mathematical model. Am Nat. 1997; 150(6):685–707. https://doi.org/10.1086/286089 PMID: 18811331
- 55. Eakin L, Wang M, Dwyer G. The effects of the avoidance of infectious hosts on infection risk in an insect-pathogen interaction. Am Nat. 2014; 185(1):100–112. https://doi.org/10.1086/678989 PMID: 25560556
- 56. Wearing HJ, Rohani P, Keeling MJ. Appropriate models for the management of infectious diseases. PLoS Med. 2005; 2(7):e174. https://doi.org/10.1371/journal.pmed.0020174 PMID: 16013892
- 57. Dwyer G, Dushoff J, Yee SH. The combined effects of pathogens and predators on insect outbreaks. Nature. 2004; 430:341–345. https://doi.org/10.1038/nature02569 PMID: 15254536
- 58. Rohrmann GF. Baculovirus Molecular Biology. Bethesda: National Library of Medicine (US); 2008.
- Sanjuán R, Domingo-Calap P. Mechanisms of viral mutation. Cell Mol Life Sci. 2016; 73(23):4433– 4448. https://doi.org/10.1007/s00018-016-2299-6 PMID: 27392606
- **60.** Meyer M, Kircher M. Illumina sequencing library preparation for highly multiplexed target capture and sequencing. Cold Spring Harb Protoc. 2010; 2010(6):1–10.
- Kuzio J, Pearson MN, Harwood SH, Funk CJ, Evans JT, Slavicek JM, et al. Sequence and analysis of the genome of a baculovirus pathogenic for *Lymantria dispar*. Virology. 1999; 253:17–34. <u>https://doi.org/10.1006/viro.1998.9469</u> PMID: 9887315
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nat Methods. 2012; 9(4):357– 359. https://doi.org/10.1038/nmeth.1923 PMID: 22388286
- **63.** Koboldt DC, Zhang Q, Larson DE, Shen D, McLellan MD, Lin L, et al. VarScan 2: somatic mutation and copy number alteration discovery in cancer by exome sequencing. Genome Res. 2012; 22(3):568–576. https://doi.org/10.1101/gr.129684.111 PMID: 22300766